

Article



# Effective Approaches to Fetal Brain Segmentation in MRI and Gestational Age Estimation by Utilizing a Multiview Deep Inception Residual Network and Radiomics

Moona Mazher <sup>1,\*</sup>, Abdul Qayyum <sup>2</sup>, Domenec Puig <sup>1</sup> and Mohamed Abdel-Nasser <sup>3</sup>

- <sup>1</sup> Departament d'Enginyeria Informatica i Matemátiques, Universitat Rovira i Virgili, 43007 Tarragona, Spain
- <sup>2</sup> School of Biomedical Engineering and Imaging Sciences, Kings College London, London SE1 9RT, UK
- <sup>3</sup> Electronics and Communication Engineering Section, Electrical Engineering Department, Aswan University, Aswan 81528, Egypt
- \* Correspondence: moona.mazher@estudiants.urv.cat

Abstract: To completely comprehend neurodevelopment in healthy and congenitally abnormal fetuses, quantitative analysis of the human fetal brain is essential. This analysis requires the use of automatic multi-tissue fetal brain segmentation techniques. This paper proposes an end-to-end automatic yet effective method for a multi-tissue fetal brain segmentation model called IRMMNET. It includes a inception residual encoder block (EB) and a dense spatial attention (DSAM) block, which facilitate the extraction of multi-scale fetal-brain-tissue-relevant information from multi-view MRI images, enhance the feature reuse, and substantially reduce the number of parameters of the segmentation model. Additionally, we propose three methods for predicting gestational age (GA)-GA prediction by using a 3D autoencoder, GA prediction using radiomics features, and GA prediction using the IRMMNET segmentation model's encoder. Our experiments were performed on a dataset of 80 pathological and non-pathological magnetic resonance fetal brain volume reconstructions across a range of gestational ages (20 to 33 weeks) that were manually segmented into seven different tissue categories. The results showed that the proposed fetal brain segmentation model achieved a Dice score of  $0.791 \pm 0.18$ , outperforming the state-of-the-art methods. The radiomics-based GA prediction methods achieved the best results (RMSE: 1.42). We also demonstrated the generalization capabilities of the proposed methods for tasks such as head and neck tumor segmentation and the prediction of patients' survival days.

Keywords: multi-view segmentation; fetal brain; fetal age prediction; deep learning; machine learning

# 1. Introduction

Congenital disorders are some of the leading causes of infant mortality worldwide [1]. Recently, in utero magnetic resonance imaging (MRI) of the fetal brain has emerged as a valuable tool for investigating the neurological development of fetuses with congenital disorders to aid in prenatal planning. Fetal MRI requires clinical and technical expertise and is a challenging imaging modality due to the ability to move freely. T2-weighted single-shot fast spin echo (ssFSE) sequences, such as ultra-fast MRI sequences, can be used to attain information in all planes.

Super-resolution (SR) reconstruction algorithms, including outlier rejection and motion correction strategies [1], can then be applied in order to combine several low-resolution images into a single high-resolution volume that can be used for further quantitative analysis. Automated quantification of the highly complex and rapidly changing brain morphology in MRI data could improve the diagnostic and decision-making processes.

Image segmentation is an early step for the volumetric quantification of the fetal brain. Shape or volume information could be relevant to the developing cortex, cerebellum, brainstem, white matter, and cerebrospinal fluid spaces [2,3]. The automatic segmentation



Citation: Mazher, M.; Qayyum, A.; Puig, D.; Abdel-Nasser, M. Effective Approaches to Fetal Brain Segmentation in MRI and Gestational Age Estimation by Utilizing a Multiview Deep Inception Residual Network and Radiomics. *Entropy* 2022, 24, 1708. https://doi.org/ 10.3390/e24121708

Academic Editors: Victor B. Kazantsev and Saikat Gochhait

Received: 30 August 2022 Accepted: 18 November 2022 Published: 23 November 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). of the developing human brain is a primary step for analysis, as manual segmentation is time-consuming and may be prone to human error. However, fetal brain segmentation based on SR fetal brain volumes is still challenging due to artifacts that are blurry or caused by motion, rapidly changing fetal brain anatomy, and the effects of partial volume.

Various atlas-based methods have been developed for brain tissue segmentation [4]. However, these methods need an atlas, which now only exists for normally developing fetuses. Falick et al. [5] used single-class high-resolution fetal brain volumes for fetal brain segmentation, but multiclass segmentation still needs to be explored. Deep-learning-based segmentation models have recently been employed to segment the fetal brain into different tissue types by using low-resolution coronal-direction slices to handle fetal brain tissue segmentation problems [6].

Faghihpirayesh et al. [7] used an encoder–decoder UNet model with multiple branches and skip connections to maintain high accuracy while devising a parallel combination of convolution and pooling operations. They used a private dataset to train their proposed model. However, they only handled the single-class segmentation problem by using 2D slices, which is not challenging and quite simple. A 2D segmentation model for volumetric 3D segmentation cannot handle temporal relationships, unlike 3D segmentation models. Moreover, they used only binary class segmentation, while the proposed model addresses the problem of multi-tissue fetal brain segmentation. Asis et al. [8] used an end-to-end generative adversarial neural network (GAN) to segment the fetal brain in functional magnetic resonance images (rs-fMRI). They segmented the full fetal brain and handled binary class problems by using a private dataset. Unlike the models in these works, the proposed multi-view segmentation model can handle the 3D segmentation of volumetric data by using a stacking approach to multi-view segmentations.

Zhao et al. [9] trained a patch-based 3D segmentation model for fetal brain segmentation by using an in-house dataset. This 3D segmentation model required powerful computational resources. However, the 3D-CNN holds great potential for fully utilizing the 3D information from MRI data, which also contain multi-view information. However, 3D-CNN-based segmentation greatly increases the network scale and computational cost [10]. It should be noted that the major bottleneck in the development of segmentation algorithms for medical imaging is the lack of data—either the availability of atlases for atlas-based segmentation or that of training data for supervised machine learning methods. In addition, there is still a need to explore and implement deep-learning-based approaches, as no clear benchmark is available for fetal brain segmentation.

In turn, the dating of the precise gestational age (GA) is essential for assessing pregnancy, fetal development, and neonatal care. Before sonography, obstetricians routinely relied on the last menstrual period for the dating of the gestational age in pre-birth life [5]. The crown-rump length (CRL) method is used in the first trimester to estimate gestational age. Other methods are used in the last two trimesters, such as brain bi-parietal diameter, head circumference, femur length, and abdominal circumference. These methods were reported decades ago and are still used today [5].

Though sonographic assessment during the first trimester is the most well known and accurate method for estimating the gestational age, it shows large variations in the second and third trimesters due to the variability in organ size. According to previous studies, the assessment of gestational age by combining the above-mentioned biometric data can achieve an accuracy of  $\pm 7$  to 10 days for the second trimester and  $\pm 21$  to 30 days for the third trimester. Various methods, such as the measurement of the cerebellar length and the transcerebellar diameter, accurately predict gestational age in singleton and twin pregnancies [11]; however, they require good visualization of the cerebellum by specialized sonographers.

In summary, estimations made with sonographic measurements are strongly affected by the inherent variability in organ size and the intrinsic signal properties of ultrasonography [12]. The inaccuracy of sonographic assessment has driven the need to find different approaches that can be used to accurately determine gestational age. MRI is gradually being recognized as a powerful helper for ultrasonography in the evaluation of the fetal brain. MRI-based methods provide a high resolution, soft-tissue contrast, and visibility of the whole brain independently of fetal presentation [13,14]. As pregnancies advance, the biological variations among normal fetuses increase, and the ranges of values of each biometric measurement associated with a specific GA also increase. This means that while the predictive error at  $\pm 10$  days GA is considered acceptable in most clinical settings, the predictive error at  $\pm 18$  days is estimated to offer little clinical value [15]. Therefore, when screening occurs in the second and third trimesters, the error margins produced by current methods are highly increased; thus, they are not clinically useful. Accordingly, there is a need to develop an alternative technique for estimating the GA.

Fung et al. [16] developed a machine learning (ML) model for estimating the GA and predicting future growth. They utilized multi-center, international, and populationbased project data from the International Fetal and Newborn Growth Consortium for the 21st Century (INTERGROWTH-21st). Kojita et al. [17] developed VGG-based transfer learning models for GA prediction. They employed an in-house (private) dataset. The deep learning model was trained with T2-weighted images from 126 training cases and 29 validation cases. The remaining 29 cases were utilized as test data, with the fetal age being estimated by the model and by using BPD (biparietal diameter) measurements. They drew a relationship between the estimated and standard gestational ages by using Lin's concordance correlation ( $\rho c$ ). The model's outcome in terms of concordance was significant ( $\rho c = 0.964$ ).

Furthermore, Lu et al. [18] developed machine learning models that could provide accurate estimations for obstetricians alongside traditional clinical practices and an efficient and effective supporting tool for pregnant women for self-monitoring. A total of 4212 intrapartum recordings were selected, of which 3370 samples were used as the training set and 842 samples were used as the test set. In addition, several simple and powerful machine learning algorithms were trained, and their performance was evaluated with real test data. The experimental results showed an intersection over union (IoU) of 0.64 between the predicted range of fetal weight at any gestational age from the ensemble model and that from ultrasound. Using their private dataset, they used simple clinical features with traditional machine learning models for the prediction of gestational age and weight. No deep-learning-based models were used as a comparison with the machine learning models. No efficient feature engineering approaches were used to predict gestational age.

Alliance et al. [19] developed a novel method based on machine learning models and used each subset of predictors based on an ensemble model constructed by using the Super Learner algorithm. The resulting model was a weighted average of multivariate adaptive regression splines, random forests, gradient boosting, support vector machines, and multiple linear regression. They assessed the diagnostic accuracy by using the receiver operating curve (AUC) and Bland–Altman analysis. They collected datasets from population-based cohorts in five countries (Bangladesh, Ghana, Pakistan, Tanzania, and Zambia). Women at <20 weeks of gestation according to ultrasound-based dating were used as a study case for the prediction of gestational age. A total of 7428 liveborn infants were included. This dataset is not publicly available. The resulting model was a weighted average of multivariate adaptive regression splines, random forests, gradient boosting, support vector machines, and multiple linear regression. They achieved the highest AUC of 0.96. They used only clinical features with traditional machine learning models for the age prediction. No imaging-based features were used to evaluate the performance of the machine learning or deep learning models for the prediction of gestational age.

Payette et al. [20] employed deep learning models such as ResNet-18 and ResNet-50 with a combination of different layers for the prediction of gestational age. They collected 741 fetal brain MRIs in order to predict fetal gestational age (in days). The authors proposed a basic ResNet18-based regressor model that used a private dataset, and they did not use any other segmentation-based or 3D volumetric-based features for gestational age prediction. They used cropped 2D images covering the fetal area only in the input images to train the

basic ResNet18 with overall global features, and this could efficiently help in the extraction of local image features for the prediction of fetal gestational age.

Shen et al. [21] used attention-guided, multi-plane ResNet-50 models trained on Stanford data to predict the gestational age. They trained various CNN models based on only imaging features for the prediction of gestational age. Imaging features might not be sufficient to accurately predict gestational age. However, we used various feature extraction approaches, including imaging, radiomics, 3D latent space autoencoder-based features, and deep features extracted from the last layer of multi-view 2D image slices from segmented brain tissues, to extract more localized features for the prediction of gestational age. The fusion of multi-scale segment-based deep features achieved better performance than that of the state-of-the-art methods.

There is a further need to investigate different methods with deep learning models for GA prediction. The existing methods are based on single-feature extraction techniques that use basic deep learning models. Correspondingly, the datasets used for the existing methods are in-house and private. There is a need to set a benchmark on a publicly available dataset for further comparisons and enhancements in deep learning/machine learning for the prediction of gestational age and segmentation of the fetal brain.

To solve the above-mentioned issues, we propose effective yet automatic methods for the segmentation of fetal brain tissue and prediction of gestational age. To the best of the authors' knowledge, this is the first paper to propose an end-to-end solution for fetal brain segmentation in MRI images and GA prediction. Deep learning is the basis for the proposed fetal segmentation method, IRMMNET (inception residual multi-scale multi-view network). By effectively combining segmentation maps from the axial, coronal, and sagittal views to create a 3D segmentation volume, IRMMNET incorporates important insights from multi-view MRI. IRMMNET consists of several layers with the capacity of reusing features and information at several scales and depths. The inception residual encoder block (EB) and the dense spatial attention (DSAM) block are two proposed blocks that are part of IRMMNET. The EB aids in extracting information from multi-view MRI scans that is pertinent to multi-scale fetal brain tissue. The DSAM improves feature reuse while lowering the model's parameter count. The EB and DSAM help segment small lesions that have a small number of semantic pixels that are missed by traditional encoder-decoder networks. Then, we propose three methods for GA prediction—GA prediction using the IRMMNET segmentation model's encoder, GA prediction using a 3D autoencoder, and GA prediction using radiomics features. The implementation of the proposed method will be available at https://github.com/Moona-Mazher/Fetal-Segmentation-Gestational-Age-Prediction-Deep-Learning on 20 November 2022.

The following is a list of this paper's major contributions:

- 1. Proposal of a novel multi-view multi-scale 3D fetal brain segmentation method named IRMMNET. It combines the key insights from multi-view MRI, including the axial, coronal, and sagittal views. IRMMNET comprises different layers with feature reuse capabilities and with various depths and multi-scale information. An efficient method for fusing segmentation maps of the axial, coronal, and sagittal views to develop a 3D segmentation volume is also presented.
- 2. Presentation of two effective blocks: the inception residual encoder block (EB) and the dense spatial attention (DSAM) block. The EB helps the fetal brain segmentation network extract multi-scale fetal-brain-tissue-relevant information from multi-view MRI images. The DSAM block enhances feature reuse and substantially reduces the number of parameters of the segmentation model. Extensive experiments were performed with various combinations and settings of the fetal brain segmentation model.
- 3. Proposal of three approaches to predicting GA: GA prediction by utilizing the IRMM-NET segmentation model's encoder, GA prediction by utilizing a 3D autoencoder, and GA prediction by utilizing radiomics features. The explainability and importance of the radiomics features are also presented.

4. Demonstration of the generalization capabilities of the proposed fetal brain segmentation and GA prediction methods on two different tasks: the segmentation of head and neck tumors and the prediction of patients' survival days.

The rest of this paper is presented as follows: Section 2 introduces the proposed multi-view multi-scale 3D fetal brain segmentation method and the proposed fetal age prediction method. Section 3 presents the datasets used in our study, the proposed fetal brain segmentation model's results, and the GA prediction method's results. Section 3.5 discusses the generalization capabilities of the proposed methods. Section 4 discusses the findings of the study and its limitations. Section 5 concludes the paper and presents the future work.

### 2. Methodology

In this section, we explain the proposed multi-view multi-scale 3D fetal brain segmentation method (Section 2.1) and the proposed fetal age prediction method (Section 2.2).

### 2.1. Proposed Multi-View Multi-Scale 3D Fetal Brain Segmentation Method

Figure 1 presents a schematic diagram of the proposed multi-view multi-scale 3D fetal brain segmentation method. The three available views—axial, coronal, and sagittal—are inputted into the proposed segmentation model to generate a 2D segmentation mask for each view. Then, the resulting segmentation masks are combined to construct a 3D segmentation map. Specifically, we stack a 2D segmentation mask of each view to form a predicted 3D segmentation mask. This process can be expressed as follows:

$$O_A = \operatorname{orgmax}(\operatorname{Sigmoid}(L_A)),$$
  

$$O_S = \operatorname{orgmax}(\operatorname{Sigmoid}(L_S)),$$
  

$$O_C = \operatorname{orgmax}(\operatorname{Sigmoid}(L_C))$$
  
(1)

$$3DV[:,:,A] = \sum_{i=1}^{N} O_{A},$$
  

$$3DV[:,S,:] = \sum_{i=1}^{N} O_{S},$$
  

$$3DV[C,:,:] = \sum_{i=1}^{N} O_{C}$$
(2)

where  $L_A$  represents linear-layer neurons using the axial view,  $L_S$  using the sagittal view, and  $L_C$  using the coronal axis view.

$$3DV[:,:,:] = 3DV[:,:,A] + 3DV[:,S,:] + 3DV[C,:,:]$$
(3)

where  $O_A$  stands for the output prediction of the proposed model when using axial slices,  $O_S$  is the prediction of the proposed model when using the sagittal view, and  $O_C$  is the output when using coronal slices; 3DV is the 3D prediction volume that is reconstructed by stacking up the predicted 2D slices from each 2D view ( $O_A$ ,  $O_s$ , and  $O_c$ ).



**Figure 1.** A schematic diagram of the proposed method for 3D fetal brain segmentation from the axial, coronal, and sagittal views.

#### 2.1.1. Multi-View Multi-Scale Segmentation Network

Figure 2 presents the proposed multi-view multi-scale segmentation network. The proposed model is designed based on the concept of an encoder–decoder with skip connections. The proposed inception residual encoder block (EB) and dense spatial attention (DSAM) block are used in the encoding path. In contrast, an efficient yet simple 2D convolutional layer module with 2D upsampling layers, including regularization layers, is used on the decoding path.



Figure 2. The proposed IRMMNET multi-view multi-scale segmentation network.

In the encoding path, a DSAM block is used in each encoder block, which sends information at every block from each encoder layer to the bottom layer. The number of channels is doubled at each EB block, and the input size of the feature maps is reduced by half based on the depth-wise convolutional layer in the encoding path. An increasing number of feature map blocks are used in each stage of encoder blocks; the number of EB blocks progressively increases at each stage of the encoder side. The first encoder block uses one EB block. Similarly, the second, third, and fourth blocks employ 2, 3, and 4 inception residual blocks, respectively.

The red lines in Figure 2 highlight the multi-scale feature maps from each encoder block fed to the model's bottom layer. This strategy increases the efficiency of the feature maps by reusing and fusing the feature information at the level of downsampling. In the proposed segmentation model, the features are extracted from three levels (red lines), which enables the model benefit from the multi-scale transformation of high-level semantic information and low-level information of the position and texture. Three downsampling layers that carry features' information are passed to the bottom-layer module to guarantee an improved cross-level feature connection and complementarity in cross-level information.

In the decoding path, the size of the feature maps increases after each 2D upsampling layer, and the original size of the training input images returns in the output in the final

layer. The first 2D upsampling layer comprises two efficient convolutional layers (2D 1  $\times$  1 Conv, 2D 3  $\times$  3 Conv) with a BN and ReLU layer. To reconstruct the semantic information, the feature maps are concatenated with each encoder and decoder block. The BN and ReLU regularization layers are used with 2D upsampled layers and 2D 3  $\times$  3 Conv layers for smooth optimization and training of the proposed model. The 1  $\times$  1 Conv layer and sigmoid activation function are used for the reconstruction of the segmentation map. Below, we explain the architectures of the proposed EB and DSAM blocks.

### 2.1.2. Inception Residual Encoder Block (Eb)

Figure 3 depicts the proposed inception residual encoder block. Unlike in the Inception-Res architecture [22], we introduce a batch normalization (BN) layer after each convolutional layer, except for the bottleneck layers. In addition, we use  $1 \times 1$  and  $1 \times 3$  kernels with a  $1 \times 5$  kernel branch, which was inspired by the DeepLab architecture [23].



Figure 3. The schematic diagram of the proposed inception residual encoder block (EB).

It should be noted that the batch normalization layer produces smooth training and can avoid gradient vanishing while retaining the convolutional layers. The feature maps are aggregated by convolving them with three kernels:  $1 \times 1$ ,  $1 \times 3$ , and  $1 \times 5$ . It is assumed that  $x_l$  is the input and  $x_{l+i}$  is the output of the *i*th layer.  $c_{1\times n}$  is a  $1 \times n$  kernel convolutional layer and  $c_b$  represents the batch normalization layer.  $c_{1\times 1}$  denotes the bottleneck layer. The output of each EB module from the encoder path can be expressed as follows:

$$x_{l+1} = l_1 \times l_2 \times l_3 \times l_4 \tag{4}$$

$$l_1 = (c_{1 \times 1}(k))c_b \tag{5}$$

$$l_2 = Maxpool(c_{1\times 1}(k)c_b) \tag{6}$$

$$l_3 = c_b(c_{1\times3}(c_b(c_{1\times3}(c_{1\times1}(k)))))$$
(7)

$$l_4 = c_b(c_{1\times 5}(c_b(c_{1\times 5}(c_{1\times 1}(k)))))$$
(8)

where  $k = x_l$ .

# 2.1.3. The Proposed Dense Spatial Attention Module (DSAM)

It should be noted that the attention modules that are often used in image segmentation and object detection models are mainly divided into channel-wise attention and point-wise attention modules, and the representatives of these two attention models are squeeze and excitation (SE) and the spatial attention module (SAM) [24]. A channel attention map exploits features' inter-channel relationships, and the feature map obtained from the channel attention is considered a feature detector. The spatial attention module focuses on what is meaningful given an input image through the benefit of the combination of average pooling and max pooling. Figure 4 shows a schematic diagram of the DSAM. As shown, we modified the SAM by replacing the point-wise convolutional layer with a basic dense layer (DL). We also used the swish activation function instead of the sigmoid function [25], as swish is less prone to vanishing gradient problems. The swish activation function can be expressed as follows:

$$Swish(x) = x.sigmoid(x)$$
(9)

where *x* is the input feature map.



**Figure 4.** A schematic diagram of the proposed dense spatial attention module (DSAM). DL is a basic dense layer. The input x is multiplied by the DL and by the swish activation function (x \* DL \* (x/(1 + e - x))).

In the DL, the feature maps of all preceding layers are used as inputs, and their feature maps are used as inputs for all subsequent layers. The output of each dense block is concatenated with every previous dense block. The operation of the DL can be expressed as follows:

$$DL = [x, B_1(x), B_2([x, B_1(x)]), B_3(x, B_1(x), B_2([x, B_1(x)])), \dots]$$
(10)

where  $B_1(x)$  is dense block 1,  $B_2$  is dense block 2, and so on. x is the input feature map. We used 12 dense blocks in our case. The output of the proposed DSAM block,  $DSAM_{out}$ , can be formulated as follows:

$$DSAM_{out} = x * DL * (x/(1+e^{-x}))$$
 (11)

The key advantages of the DL are (1) the alleviation of the vanishing gradient problem, (2) the strengthening of feature propagation, (3) the encouragement of feature reuse, and (4) the substantial reduction in the number of parameters. The feature maps are concatenated from the previous layer to the next layer to build the dense block in the proposed dense layer. The feature maps keep the relevant information from every layer and are reused in the final layer to get semantic information. The proposed DSAM uses the dense block to provide better semantic information and improves the flow of gradient information for easy training of the proposed model. The DSAM block reduces the problem of overfitting with smaller training set sizes by using dense connections. DSAM blocks also provide direct access to the gradients from the loss function and the original input signal, which leads to implicit deep supervision. It is worth noting that all layer weights in the proposed model were trained from scratch with the FeTA dataset.

### 2.1.4. Loss Function and Implementation Details

In this paper, we employ the Combo loss function [26] to train the proposed model with multi-class settings for fetal brain tissue segmentation. Combo loss can be expressed as follows:

$$L = \alpha \left( -\frac{1}{N} \sum_{i=1}^{N} \beta(t_i - \ln p_i) + (1 - \beta) [(1 - t_i) \ln(1 - p_i)] \right) - (1)$$
  
-\alpha) 
$$\sum_{i=1}^{K} \left( \frac{2\sum_{i=1}^{N} p_i t_i + S}{\sum_{i=1}^{N} p_i + \sum_{i=1}^{N} t_i + S} \right)$$
(12)

where  $t_i$  is the one-hot-encoded target or ground truth,  $p_i$  is the predicted probability, N is the number of classes multiplied by the number of samples, and the S is a small constant number that is added to prevent division by zero.  $\alpha$  controls the amount contributed by the Dice term in the loss function L.  $\beta \in [0, 1]$  controls the level of penalization of the model for false positives/negatives. The S term is added to prevent division by zero. The S constant is added in both the denominator and the numerator of the Dice term.

All models were trained using an Adam optimizer with a learning rate of 0.0001,  $\rho = 0.95$ ,  $\epsilon = 1 * 10^{-8}$ , and decay = 0. Based on the experimental evaluations, it was found that  $\alpha = 0.5$  for the Dice and cross-entropy terms produced the best results. Different values of  $\beta$  were tried for all datasets, and it was found that  $\beta = 0.5$  was the best value for the proposed dataset for the segmentation task.

### 2.2. Approaches to the Prediction of Gestational Age (GA)

This subsection proposes three approaches to predicting GA: (1) GA prediction by utilizing the IRMMNET segmentation model's encoder, (2) GA prediction by utilizing a 3D autoencoder, and (3) GA prediction by utilizing radiomics features.

### 2.2.1. GA Prediction by Utilizing the IRMMNET Segmentation Model's Encoder

The encoder of the proposed IRMMNET segmentation model can automatically learn multiple filters in parallel and extract low- and high-level features, such as edges, intensities, and textures. Different filters capture various characteristics of the input images that are used in GA prediction. Figure 5 presents the proposed framework for GA prediction by utilizing the IRMMNET segmentation model's encoder. As one can see, deep features are extracted from the trained encoder of each view's trained segmentation model. Each view's encoder generates a feature vector with dimensions of  $1 \times 256$ . It should be noted that the feature vectors of all slices of each volume are combined to produce one feature vector that represents the whole volume. The feature vectors of the three views are concatenated to form one single feature vector with dimensions of  $1 \times 768$ . Different regression algorithms are trained with the extracted feature vector in order to predict the GA.

### 2.2.2. GA Prediction by Utilizing a 3D Autoencoder

Figure 6 presents the proposed approach to GA prediction by utilizing a 3D autoencoder. The 3D autoencoder consists of an encoder- and decoder-based model. A 3D volume is fed to the proposed 3D autoencoder to extract latent space features. As shown in Figure 6, the encoder part of the 3D autoencoder consists of a 3D convolutional layer, batch normalization layer, ReLU layer, and 3D maxpool layer, and the decoder side of the 3D autoencoder comprises a 3D transposed layer at each level on that side along, with a 3D convolutional layer and a batch normalization layer.

On the encoder side, the volume is reduced at each encoder block, while the volume is increased at each decoder block. The 3D transposed layer is used to upsample the feature maps at each decoder block. The bottom layer represents the 3D volume in the lower dimension. After flattening the lower dimension, a latent vector is produced. Different regression algorithms are trained with the latent features to predict the GA, as shown in Figure 6.





Figure 5. GA prediction by utilizing the IRMMNET segmentation model's encoder.



**Figure 6.** GA prediction by utilizing a 3D autoencoder. The latent features are extracted by using input image volumes for GA prediction.

# 2.2.3. GA Prediction by Utilizing Radiomics Features

Figure 7 shows the proposed approach to GA prediction by utilizing radiomics features. Here, we also extract 2D slices of the axial, sagittal, and coronal views from a 3D volume dataset. We train the proposed segmentation model for the dataset containing the three views and stack the output of each 2D view to reconstruct a 3D volume. The radiomics features are extracted from the input volumes of the dataset. The radiomics features used include shape-based, statistical, and wavelet features. Among the 108 radiomics features, a set was chosen based on a correlation-based feature selection technique [27]. The selected radiomics features were elongation, flatness, major axis length, minor axis length, max 3D diameter, sphericity, surface area, energy, entropy, kurtosis, mean, skewness, coarseness, contrast, correlation, inverse-diff moment, complexity, and strength. The radiomic features were extracted from Feta MRI images by using the Pyradiomics package [28]. Different regression algorithms were trained with the extracted radiomics features to predict the GA.

# 2.2.4. Regression Techniques for GA Prediction

Different regression techniques were tested on the features extracted with the three approaches mentioned above for the prediction of the GA. We found that four regression techniques give acceptable results: random forest (RF) [29], regression trees (RT), linear regression (LR), and extreme gradient boosting (XGB) [30]. RF is supervised by traditional machine learning and is widely used for classification and regression problems. It uses a bootstrapped dataset as a subset, picks random subsets of features, and runs random trees in parallel while building the trees. We set the number of trees to be from 100 to 1000 to create

the forest. LR is a linear model that builds a linear relationship between input variables and a single output variable. RT is a tree-based regression model that trains a model by observing the input object's features and generating a continuous output. The gradientboosting regressor is a method that uses an additive forward model and allows arbitrary differentiable loss functions for optimization during training. The gradient-boosting model is an ensemble model that can be used for regression, classification, and predictive modeling problems. Extreme gradient boosting (XGB) is an open-source approach to gradientboosting regression.



Figure 7. GA prediction by utilizing radiomics features.

# 3. Experimental Results and Discussion

In this section, we explain the dataset used in our study (Section 3.1), present and discuss the results of the proposed fetal brain segmentation model (Section 3.3), and analyze the results of the GA prediction models (Section 3.4).

# 3.1. Dataset Description

The dataset included 80 T2-weighted fetal brain reconstructions with a corresponding label map that was manually segmented into seven different tissues/labels [20]. The seven labels were external cerebrospinal fluid (ECF), fluid gray matter (FGM), white matter (WM), ventricles (VCs), cerebellum (CBM), deep gray matter (DGM), and brainstem (BSTM). The dataset consisted of clinically acquired fetal brain reconstructions of both neurotypical and pathological brains with a range of gestational ages. The data were acquired using 1.5*T* and 3*T* clinical GE whole-body scanners (Signa Discovery MR450 and MR750) with either an eight-channel cardiac coil or a body coil. T2-weighted single-shot fast spin echo sequences were acquired with an in-plane resolution of 0.5 mm × 0.5 mm and a slice thickness of 3 to 5 mm. The sequence parameters were the following: TR: 2000–3500 ms; TE: 120 ms (minimum); flip angle: 90°; sampling percentages: 55%.

Figure 8 shows the class-mapping function for the axial, sagittal, and coronal slices. The different colors show the seven classes used to predict fetal tissue segmentation.

Two different methods were used to create a super-resolution reconstruction of the fetal brain for each case from the acquired low-resolution axial, coronal, and sagittal images. Equal numbers of cases were used in the training and the evaluation datasets for each reconstruction method. For each case, the gestational age in weeks and the neurotypical/pathological label was given, in addition to the label maps.

The dataset was divided into 80% for training and 20% for testing. There were totals of 64 subjects in training and 16 subjects in the testing phase.



Figure 8. The class-mapping function for the axial, sagittal, and coronal slices.

# 3.2. Evaluation Metrics

In this study, four metrics were used to assess the performance of the fetal brain segmentation models: Dice (DSC), Hausdorff distance (HD95), sensitivity, and specificity, which are are commonly used for the validation of medical volume segmentation approaches [31]. This is also called the overlap index. It measures the overlap between ground-truth (GT) and predicted segmentation masks. For the GT and predicted masks, DSC is defined as follows:

$$Dice(X,Y) = 2|X \cap Y| / |X \cup Y|$$
(13)

Hausdorff distance (HD95): The HD95 is calculated as the mean of two directed 95% Hausdorff distances:

$$HD95 = \frac{\vec{d}H,95(X,Y) + \vec{d}H,95(Y,X)}{2}$$
(14)

where X is the ground truth (GT) and Y is the predicted mask. HD is the maximum distance between the sets of points X and Y and between Y and X.

Sensitivity is used to compute the positive portion of voxels by using the ground-truth (GT) and predicted segmentation masks.

Sensitivity 
$$= TRP = TP/(TP + FN)$$
 (15)

where TRP is the true positive rate, TP is true positive, and FN is false negative.

Specificity is also called the true negative rate (TNR), and it is used to compute performance based on the GT and predicted segmentation masks.

Specificity 
$$= TNR = TN/(TN + FP)$$
 (16)

where TP is true positive, FP is false positive, TN is true negative, and FN is false negative.

In addition, we use the root-mean-square error (RMSE) and concordance (C-index) to evaluate the GA prediction models. The C-index is used to compute the correlation between the predicted gestational age and ground-truth gestational age. The RMSE can be expressed as follows:

$$RMSE = \sqrt{\frac{\sum (GA_{pre} - GA_{GT})^2}{N}}$$
(17)

The C-index can be formulated as follows:

$$C-index = concordance - index(GA_{GT}, GA_{pre})$$
(18)

where  $GA_{pre}$  is the predicted value,  $GA_{GT}$  is the ground-truth value for the *i*th observation in the dataset, and N is the sample size.

# 3.3. Performance Analysis of the Proposed Segmentation Model

### 3.3.1. Ablation Study

Table 1 tabulates the DSC, HD95, sensitivity, and specificity values of the proposed segmentation model, IRMMNET, with the axial, sagittal, and coronal views (IRMMNET-Axial, IRMMNET-Sagittal, and IRMMNET-Coronal). IRMMNET-Coronal obtained better performance for all classes than IRMMNET with the axial and sagittal planes. It achieved DSC, HD95, sensitivity, and specificity scores of 0.789, 21.56, 0.818, and 0.976, respectively.

Table 1. Performance of the proposed segmentation model with the axial, sagittal, and coronal views.

Model	DSC	HD95	Sensitivity	Specificity
IRMMNET_Axial	0.778	24.06	0.8126	0.974
IRMMNET _Sagittal	0.781	22.80	0.817	0.974
IRMMNET _Coronal	0.789	21.56	0.818	0.976

Figure 9 shows the proposed IRMMNET model's segmentation maps for the axial, sagittal, and coronal planes. In addition, 2D and 3D volumetric views of the segmentation images are shown in Figure 9 for the axial, sagittal, and coronal planes. It can be visibly noticed that the coronal-view model generated accurate segmentation masks, in which the predictions for the deeper and smaller classes were close to the GT. It should be noted that the proposed model was initially tried on the 2D axial slices, but the predicted segmented images yielded a bad prediction for the deep classes. Therefore, the proposed segmentation model was applied to the three views' 2D slices (axial, sagittal, and coronal) for fetal brain segmentation. Among the predicted segmentation results for the fetal brain, the 2D coronal view produced the best results.



Figure 9. The segmentation maps of the proposed model with different views.

In Table 2, we compare the proposed model with different segmentation models. Specifically, the basic UNet (BaseUNet) for 2D brain tissue segmentation was trained by using the axial, coronal, and sagittal views (BaseUNet-Axial, BaseUNet-Sagittal, and BaseUNet-Coronal). As one can see, BaseUNet-Coronal yielded the highest DSC score (0.728) and the lowest HD95 score (29.042). In addition, different ResUnet models were trained by using the axial, coronal, and sagittal views (ResUnet-Axial, ResUnet-Sagittal,

and ResUnet-Coronal) for 2D brain tissue segmentation. In the ResUnet model, the residual blocks were added to the base UNet model. The results of the three ResUnet models were better than those of the BaseUNet models. Finally, we added squeeze-and-excitation (SE) blocks into ResUNET (SE-ResUNet) and trained it on the three views, yielding SE-ResUNet-Axial, SE-ResUNet-Sagittal, and SE-ResUNet-Coronal. However, SE-ResUNet-Coronal had an improved DSC score compared to that of ResUnet-Coronal; its HD95 and specificity values were worse. As shown in Table 2, the proposed IRMMNET model comparatively produced a better performance with the axial and sagittal views. However, IRMMNET-Coronal achieved a DSC score of 0.789 and HD95 score of 21.565, which were better than those of all models used for comparison.

Model	DSC (%)	HD95 (mm <sup>3</sup> )	Sensitivity (%)	Specificity (%)
BaseUNet-Axial	0.727	29.401	0.806	0.952
BaseUNet-Sagittal	0.723	31.381	0.790	0.947
BaseUNet-Coronal	0.728	29.042	0.813	0.963
ResUNet-Axial	0.748	26.111	0.827	0.968
ResUNet-Sagittal	0.756	27.334	0.818	0.969
ResUNet-Coronal	0.752	26.014	0.828	0.977
SE-ResUNet-Axial	0.762	28.262	0.809	0.973
SE-ResUNet-Sagittal	0.769	28.888	0.808	0.978
SE-ResUNet-Coronal	0.773	27.101	0.879	0.969
IRMMNET-Axial	0.778	24.062	0.819	0.972
IRMMNET-Sagittal	0.781	22.801	0.817	0.974
IRMMNET-Coronal	0.789	21.565	0.818	0.976

Table 2. Comparing IRMMNET with different segmentation models.

To enhance the proposed model's prediction, we fused the axial, sagittal, and coronal outputs of the proposed model to create a so-called multi-view model, which provided a 3D segmentation map. Later, the performance was evaluated by using the predicted 3D segmentation map achieved with our three multi-view models and a GT segmentation map. It resulted in a better estimation in terms of the Dice, HD95, sensitivity, and specificity scores. We constructed a 3D segmentation map from the three views of the proposed model and evaluated the performance by using the predicted 3D segmentation map (achieved with our three multi-view models) and ground-truth segmentation map. Table 3 presents an ablation study of the proposed IRMMNET. Although the baseline Multiview-2D-Inception+Residual model achieved optimal performance, its performance was upgraded when the DSAM module (i.e., Multi-view-2D Inception + Residual + DSAM) was systematically added. However, adding the multi-scale feature approach to the Multi-view-2D-Inception+Residual+DSAM model with the fusion of multiple views (i.e., IRMMNET) produced the highest performance scores in comparison with those of all of the state-of-artmethods and the baseline model for the fetal brain segmentation task when using the FeTA 2021 dataset. IRMMNET achieved DSC, HD95, sensitivity, and specificity scores of 0.791, 21.66, 0.819, and 0.980, respectively.

Table 3. Performance of various configurations of the proposed IRMMNET model.

Model	DSC	HD95	Sensitivity	Specificity
UNet	0.733	28.58	0.817	0.968
Multi-view_2D Inception + Residual	0.778	25.42	0.8178	0.967
Multi-view_2D Inception + Residual + DSAM	0.783	23.26	0.8101	0.976
IRMMNET	0.791	21.66	0.819	0.980

We also studied the efficacy of different loss functions with the proposed model. As tabulated in Table 4, the proposed Combo loss function achieved better performance than that of the binary cross-entropy (BCE) and Dice loss functions. No big improvements were

**Loss Function** DSC (%) HD95 (mm<sup>3</sup>) Sensitivity (%) Specificity (%) BCE 0.789 23.88 0.671 0.812 0.668 0.795 Dice 0.77624.51BCE + Dice 0.780 22.83 0.682 0.809 0.691 0.791 Combo 21.66 0.818

noticed when we combined the Dice loss and BCE loss. However, the Dice loss produced the lowest scores.

Table 4. Performance of IRMMNET with different loss functions.

computational times for training and validation are given in Table 5.

The training and validation times were also estimated for the proposed and the stateof-the-art methods for a comparison of the computational costs. The training time of our proposed solution was 55 min, and the time taken for validation was less than 2 min. The

Table 5. Estimations of the training and validation times for the proposed and state-of-the-art methods.

Segmentation Models	Training Time (min)	Validation Time (min)
BaseUNet-Axial	45	2
BaseUNet-Sagittal	50	2.1
BaseUNet-Coronal	55	2.01
ResUNet-Axial	60	2.23
ResUNet-Sagittal	62	2.45
ResUNet-Coronal	61	2.11
SE-ResUNet-Axial	63	2.53
SE-ResUNet-Sagittal	65	1.95
SE-ResUNet-Coronal	64	1.88
IRMMNET-Axial	53	1.3
IRMMNET-Sagittal	55	1.4
IRMMNET-Coronal	52	1.5

We applied a Mann–Whitney U test or Wilcoxon Rank Sum test to compute the *p*-values between the predicted masks and their corresponding ground truths [32,33]. In segmentation tasks, the *p*-value needs to be higher than 0.05 to be statistically significant, unlike in classification tasks. A comparison of the statistical analyses of the proposed IRMMNET and the state-of-the-art methods is given in Table 6. In the table, a *p*-value that is greater than 0.05 represents a greater similarity between the predicted and ground-truth segmentation maps. Similarly, a higher *p*-value also represents an accurate segmentation, and vice-versa. The table shows that the proposed IRMMNET had consistent results, and it statistically validates the segmentation results.

To validate the results of the proposed fused model (i.e., Multi-view-IRMMNET), we also applied the same fusion technique to the ResUNet and SE-ResUNet models, which yielded Multi-view-ResUNET and Multi-view-SE-ResUNET. The predicted 2D slices and 3D volumes of the proposed model, ResUnet, and SE-ResUNet are shown in Figure 10. Although ResUNet and SE-ResUNet successfully predicted outer classes, such as external cerebrospinal fluid (ECF), they failed to predict inner/deeper classes, like deep gray matter (DGM). It is conspicuous that the proposed model's predictions for all classes were close to the given GT.

Segmentation Models	<i>p</i> -Value
BaseUNet-Axial	0.65
BaseUNet-Sagittal	0.68
BaseUNet-Coronal	0.72
ResUNet-Axial	0.89
ResUNet-Sagittal	0.87
ResUNet-Coronal	0.97
SE-ResUNet-Axial	1.15
SE-ResUNet-Sagittal	1.10
SE-ResUNet-Coronal	1.01
IRMMNET-Axial	1.63
IRMMNET-Sagittal	1.7
IRMMNET-Coronal	1.9

 Table 6. Comparison of the statistical analyses of the proposed IRMMNET and the state-of-the-art methods.



**Figure 10.** The segmentation results for the proposed Multi-view-IRMMNET, Multi-view-ResUNET, and Multi-view-SE-ResUNET models.

Table 7 presents the Dice, HD95, sensitivity, and specificity scores of the proposed Multi-view-IRMMNET, Multi-view-ResUNET, and Multi-view-SE-ResUNET models. The proposed Multi-view-IRMMNET model achieved the best segmentation results with a DSC of 0.791, HD95 of 21.66, sensitivity of 0.819, and Specificity of 0.980. The Multi-view-ResUNET model achieved less performance scores than those of Multi-view-IRMMNET. This model achieved a DSC of 0.758, and Multi-view-SE-ResUNET achieved a maximum DSC of 0.772. We can conclude that our proposed Multi-view-IRMMNET achieved the highest DSC score in comparison with the other models, namely, Multi-view-ResUNET and Multi-view-SE-ResUNET. Similarly, Multi-view-IRMMNET achieved the lowest HD95 score of 21.66, as compared to those of Multi-view-ResUNET and Multi-view-SE-ResUNET. Figure 11 shows the box plots of the DSC, HD95, sensitivity, and specificity scores of the proposed Multi-view-IRMMNET, Multi-view-ResUNET, and Multi-view-SE-ResUNET models. Multi-view-IRMMNET showed the highest Q3 quartile in the DSC and specificity plots. In contrast, Multi-view-IRMMNET's median and maximum values remained higher in all plots compared to those of the Multi-view-ResUNET and Multi-view-SE-ResUNET models.

Model	Dice	HD95	Sensitivity	Specificity
Multi-view_ResUNET	0.758	27.39	0.826	0.97
Multi-view_SE-ResUNET	0.772	27.56	0.812	0.973
Multi-view_IRMMNET	0.791	21.66	0.819	0.980

**Table 7.** Comparing the segmentation results of the proposed Multi-view-IRMMNET, Multi-view-ResUNET, and Multi-view-SE-ResUNET models.



**Figure 11.** Box plots of the DSC, HD95, sensitivity, and specificity scores of the proposed Multi-view-IRMMNET, Multi-view-ResUNET, and Multi-view-SE-ResUNET models.

Figure 12 overlays the density plots of the predicted 2D DSCs of the proposed Multiview-IRMMNET, Multi-view-ResUNET, and Multi-view-SE-ResUNET models for each class. First, the distribution of the predicted 2D DSCs for Multi-view-IRMMNET was always significantly different from those of the Multi-view-ResUNET and Multi-view-SE-ResUNET models for all classes. This was especially the case for VC, CBM, DGM, and BSTM, which were the classes opposite to EFC, FGM, and WM. It is worth noting that when a class was absent in a segmented slice, the predicted 2D DSC was zero for that slice. This showed a small trend of zero appearing for all classes, especially for deep classes, such as VC, CBM, DGM, and BSTM.

Similarly, the distribution scores of all classes with Multi-view-IRMMNET were greater than those with Multi-view-ResUNET and Multi-view-SE-ResUNET. However, regardless of the class, the 2D DSC distributions of the Multi-view-ResUNET and Multi-view-SE-ResUNET models were always significantly different, were more shifted to the left, and had a larger standard deviation compared to that of the distribution of the proposed solution. The proposed solution was always shifted toward the higher values on the right.

# 3.3.2. Comparing the Proposed Segmentation Model with Existing Methods

There has not been much research on the FeTA dataset and fetal brain segmentation, as private datasets are most commonly used in research. Table 8 compares the performance of the proposed Multi-view-IRMMNET with that of two state-of-the-art methods called DA\_FaBiAN\_Baseline [34] and TopoCP (2D) [35] in terms of DSC scores. As shown, Multi-view-IRMMNET achieved the best results, with a DSC score of 0.791. However, DA\_FaBiAN\_Baseline and TopoCP (2D) produced the same DSC score (0.70), but TopoCP (2D) obtained the smallest standard deviation.



**Figure 12.** Density plots of the predicted 2D DSC for the proposed Multi-view-IRMMNET, Multi-view-ResUNET, and Multi-view-SE-ResUNET models.

**Table 8.** Comparison of the performance of the proposed Multi-view-IRMMNET model with that of two existing methods—DA\_FaBiAN\_Baseline and TopoCP (2D).

Model	DSC (%)	
DA_FaBiAN_Baseline [34] TopoCP (2D) [35] Multi-view-IRMMNET	$egin{array}{c} 0.70 \pm 0.24 \ 0.70 \pm 0.14 \ 0.791 \pm 0.18 \end{array}$	

# 3.4. GA Prediction Results

3.4.1. Analyzing the Performance of GA Prediction Models

As mentioned in Section 2.2, three different methods were used to extract features from the MRI images of the fetal brain, and then these features were fed into a regression algorithm to predict the GA. Four different regression techniques were used for the regres-

sion of the input features: LR, XGB, RF, and RT. The first feature set, which was called IRMMNET Deepfeat, was extracted from the last encoder layer of the proposed IRMMNET model, as depicted in Figure 5. The second feature set, the so-called 3D deep autoencoder features, was extracted from the autoencoder depicted in Figure 6. The third feature set included radiomics features (Figure 7).

Table 9 shows that the RF regressor produced the lowest RMSE values with the radiomics, IRMMNET Deepfeat, and 3D deep autoencoder features. The RF regressor achieved the best GA prediction results with the radiomics features, with an RMSE score of 1.42 and a C-index score of 0.888. These results indicate that the proposed radiomics features used with the RF regressor were the most reliable method for predicting GA.

**Table 9.** Analyzing the performance of different feature types and regression techniques for GA prediction.

Madala		LR	x	GB		RF		RT
Models	RMSE	C-Index	RMSE	C-Index	RMSE	C-Index	RMSE	C-Index
Radiomics	1.70	0.837	1.477	0.854	1.42	0.888	1.44	0.858
IRMMNET Deepfeat	4.56	0.465	4.10	0.252	3.46	0.371	5.97	0.542
3D deep autoencoder features	4.51	0.418	3.71	0.427	3.26	0.512	4.64	0.517

Figure 13 presents Kaplan–Meier plots of the radiomics, IRMMNET Deepfeat, and 3D deep autoencoder features. These plots show the gestational days predicted by the proposed GA prediction model in comparison with the ground truth for the validation datasets. Figure 13a shows the days predicted with the radiomics features with RF, Figure 13b shows the days predicted with the IRMMNET Deepfeat with RF, and Figure 13c shows the days predicted with the 3D deep autoencoder features with RF. As shown, the curves of the predicted days were very close to the GT curve for the radiomics-based features, unlike the curves for the IRMMNET Deepfeat and 3D deep autoencoder features. The curve of the predicted days that was produced based on the 3D deep autoencoder features was far from the curve based on the GT days. Hence, these curves prove the efficacy of the proposed GA prediction method based on radiomics features and RF.



**Figure 13.** Kaplan–Meier plots of the radiomics, IRMMNET Deepfeat, and 3D deep autoencoder features with RF.

# 3.4.2. Analyzing the Explainability of the Radiomics Features

As shown above, the radiomics features yielded the best GA prediction results. Here, we employ the SHAP explainability method (SHapley Additive exPlanations) in order to analyze the most explainable radiomics features and their importance. SHAP is a game-theoretic approach to explaining the output of any machine learning model [36]. Figure 14 shows the feature importance of the radiomics features for the further analysis of the approach to using the radiomics features for GA prediction. We used the 18 best-explainable radiomics features. It was shown that the max 3D diameter and major axis length had the highest feature importance. We can say that the max 3D diameter and minor axis length were the most important features for GA prediction in comparison with the other features, which can help in future planning.



**Figure 14.** Feature importance of the radiomics features for GA prediction. The colors from red to blue represent the significance of the features in model prediction.

# 3.5. Generalization Capabilities of the Proposed Fetal Brain Segmentation and GA Prediction Models

To demonstrate the generalization capabilities of the proposed fetal brain segmentation model, we tested it on the Hecktor 2021 dataset [37]. A total of 224 training samples were provided. This dataset also contained clinical values and imaging samples. The dataset was divided into 80% for training and 20% for testing. The Hecktor 2021 dataset was converted into axial, coronal, and sagittal views. We trained the proposed IRMMNET fetal brain segmentation model with the axial, coronal, and sagittal views from a head and neck tumor dataset. Each volume in the Hecktor 2021 dataset had a spatial resolution of  $144 \times 144 \times 144$ . The segmentation masks of the IRMMNET model for each view were fused to construct the 3D segmentation of the tumors in the head and neck dataset. Table 10 compares the results of the proposed model with those of existing state-of-the-art methods in terms of the DSC and HD95 scores. IRMMNET achieved a DSC score of 0.77 and an HD95 score of 3.02, which were better than those of the methods used for the comparison.

In addition, the proposed GA prediction method was tested on the the Hecktor 2021 dataset for the prediction of the patients' survival days. The progression-free survival outcomes for all patients were provided in the CSV files with their clinical data and with various clinical variables. The head and neck tumor progression was based on the RECIST

criteria: either an increased size of a known tumor (change in N and/or T) or the existence of a new tumor (change in M and/or N). Death due to a specific disease was also considered the progression of a disease that was previously considered stable.

**Table 10.** Analysis of the generalization capabilities of the proposed fetal brain segmentation model on the Hecktor 2021 dataset.

Method	DSC (%)	HD95 (mm)
IRMMNET	0.77	3.02
[37]	0.65	4.07
[38]	0.75	3.27

Clinical variables, such as the patient age, patient gender, center ID, TNM group, M-stage, N-stage, T-stage, TNM edition, and chemotherapy status, were given with different values, and some variables had missing values. We used imputation to complete the missing values for all clinical features. We mapped integer values to each the individual N-, M-, and T- staging datum as follows: T-stage (Tx: 0, T1: 1, T2 : 2, T3: 3, T4: 4, T4a: 5, T4b: 6), N-stage (N0: 0, N1: 1, N2:N2a: 3, N2b: 4, N2c: 5, N3: 6), and M-stage (Mx: 0, M0: 0, M1:1). In addition, the TNM group was also mapped to an ordinal categorical variable, which was based on the corresponding TNM stage information (7 I: 0, 8 I: 0, 7 II: 1, 8 II: 1, 7 III: 2, 8 III: 2, 7 IV: 3, 8 IV: 3, 7 IVA: 4, 8 IVA: 4, 7 IVB: 5, 8 IVB: 5, 7 IVC: 6, 8 IVC: 6).

The min/max normalization method provided by the scikit-learn Python package was used to normalize the clinical features' values. A scaler was instantiated by using only the training data and then applied to the test set.

In these experiments, the RF regression technique was employed. As tabulated in Table 11, the clinical features obtained a C-index of 0.692. However, the C-index of the clinical features was lower than those of the radiomics, DeepFeat (deep features extracted from the encoder of the proposed IRMMNET model), and 3D deep autoencoder features, but it was better than those in existing studies, such as [39,40]. The Clinical + DeepFeat + Radiomics combination led to a C-index of 0.786, which was lower than the Deep-Features + Radiomics combination, meaning that we could achieve accurate predictions of patients' survival days without employing clinical data. Table 11 also demonstrated that the features based on radiomics and DeepFeat achieved the highest C-index scores (0.821) compared to those of methods that were specially designed for the prediction of patients' survival days, such as [38–41].

**Table 11.** Analysis of the generalization capabilities of the proposed feature extraction methods on the Hecktor 2021 dataset for the prediction of patients' survival days.

Method	C-Index	
Clinical features	0.692	
Radiomics	0.791	
DeepFeat	0.771	
3D deep autoencoder features	0.723	
Deep-Features + Radiomics	0.821	
Clinical + DeepFeat + Radiomics	0.786	
[38]	0.810	
[39]	0.470	
[40]	0.690	

# 4. Discussion

Image segmentation is the first stage in the volumetric quantification of the developing fetal brain, which is used to examine the neurological growth of fetuses with congenital abnormalities and inform prenatal planning. The predicted delivery date is crucial for an accurate estimation of gestational age when managing any pregnancy. The timing of appropriate obstetric treatment and the scheduling and interpretation of some antepartum diagnostics that assess the appropriateness of fetal growth and measures of development in order to prevent preterm births and associated morbidities depend heavily on accurate knowledge of the gestational age. We must create an automated, accurate, and precise procedure for fetal brain segmentation and gestational age calculation.

This paper tackled the tasks of fetal tissue segmentation and gestational age prediction. The 2D-based multi-view (axial, coronal, and sagittal) models were analyzed for fetal brain tissue segmentation. The end-to-end fetal tissue segmentation and GA prediction models were trained and tested with the FeTA 2021 MRI dataset. Initially, we trained our segmentation model on only axial slices. However, the results were not convincing, especially in brain tissues in which the number of class pixels was small. In addition, we acquired all axial, sagittal, and coronal slices from the 3D fetal brain input volume. We trained the model on those particular view slices. The coronal view model performed better than the sagittal and axial models based on the experimental results and observations. Later, the fusion of the predictions of the axial, sagittal, and coronal views were combined to enhance the model's outcome.

Various regression techniques were developed for GA prediction by using different feature extraction methods, including a 3D autoencoder, radiomics features, and features extracted from the IRMMNET encoder. The IRMMNET encoder's features were used with different regressors (LR, RF, RT, and XGB) to predict gestational age. Radiomics- and 3D-autoencoder-based features were also used for GA prediction. Extensive experiments were performed to achieve the optimal performance in optimal segmentation and gestational age prediction. Metrics such as those of the Kaplan–Meier and the SHAP explainability methods were used to study the explainability of the proposed GA models. Different comparisons and datasets were utilized to validate the generalization of the proposed models.

The main limitation of our study is that the sample size for each GA period in the held-out validation dataset was relatively small. Therefore, validating the proposed GA estimation model with data from larger populations and settings will be critical in order to extend the current use of this MRI-based biometric measurement prediction model to clinical application scenarios. Since fetal growth is influenced by each mother's previous gestational history, body condition, and composition, future studies should consider mothers' demographics as variables or covariables in their models, which might be useful for improving the precision of gestational dating. We chose multi-view multi-scale deep learning segmentation models in order to deliver the complete volumetric information and to provide an efficient application to fetal brain segmentation, gestational age prediction, and head and neck tumor segmentation and survival analysis. Our proposed approach consisted of various modules for predicting 3D segmentation maps with a limited dataset and lower computational resources. Our proposed approach produced a more efficient and faster response on the validation dataset for the task of fetal brain segmentation. The imaging features might not be sufficient to accurately predict gestational age. However, we used various feature extraction approaches, which included imaging, radiomics, 3D latent space autoencoder-based features, and deep features extracted from the last layer of multi-view 2D image slices from segmented brain tissues, to extract more localized features for gestational age prediction. The fusion of multi-scale segment-based deep features achieved better performance than that of state-of-the-art methods. There is a further need to investigate different methods with deep learning models for GA prediction. The existing methods are based on single-feature extraction techniques that use basic deep learning models. Correspondingly, the datasets used in existing methods are in-house and private. There is a need to set a benchmark on a publicly available dataset for further comparisons and enhancements in deep learning/machine learning for gestational age prediction and fetal brain segmentation.

We chose multi-view multi-scale deep learning segmentation models in order to deliver the complete volumetric information and to provide an efficient application to fetal brain segmentation, gestational age prediction, and head and neck tumor segmentation and survival analysis. Our proposed approach consisted of various modules for predicting 3D segmentation maps with a limited dataset and lower computational resources. Our proposed approach produced a more efficient and faster response on the validation dataset for the task of fetal brain segmentation. We developed a couple of imaging-based features and validated them on two different medical imaging datasets for segmentation, fetal age prediction, and head and neck survival analysis. This is a comprehensive end-to-end solution for fetal brain segmentation and gestational age prediction. Three types of features—radiomics features, clinical features, and latent features from the 3D autoencoder—were used for the prediction of gestational age with various regression models. These regression models were applied together with various feature fusion combinations for the prediction of gestation age. Our proposed model was also used in head and neck cancer segmentation. The head and neck segmentation features were used in the prediction of the survival age, and the performance of the proposed model was compared with the performance of state-of-the-art models for the tasks of head and neck cancer segmentation and survival prediction. We developed a couple of imaging-based features and validated them on two different medical imaging datasets for segmentation, fetal age prediction, and head and neck survival analysis. This is a comprehensive end-to-end solution for fetal brain segmentation and gestational age prediction. Three types of features—radiomics features, clinical features, and latent features from the 3D autoencoder—were used for the prediction of gestational age with various regression models. These regression models were applied together with various feature fusion combinations for the prediction of gestation age. Our proposed model was also used in head and neck cancer segmentation. The head and neck segmentation features were used in the prediction of the survival age, and the performance of the proposed model was compared with the performance of state-of-the-art models for the tasks of head and neck cancer segmentation and survival prediction.

### 5. Conclusions and Future Work

This paper proposed an end-to-end, fully automated, and effective method for multitissue fetal brain segmentation called IRMMNET. It is a multi-view segmentation model with significantly fewer parameters thanks to the inception residual encoder block (EB) and the dense spatial attention (DSAM) block, making it easier to extract information from multi-view MRI images that is pertinent to multi-scale fetal brain tissue and to improve feature reuse. In addition, this paper presented three approaches to the estimation of gestational age (GA). The first approach was based on a 3D autoencoder, the second approach was based on radiomics features, and the third approach was based on the IRMMNET segmentation model's encoder.

Extensive experiments and analyses were provided, and these included the study of different configurations of the proposed segmentation model and the analysis of the impacts of various loss functions. We also examined the effect of applying the proposed fusion technique to other existing segmentation models, such as the UNet, ResUNet, and SE-ResUNet models. We found that the proposed fetal brain segmentation model obtained the best results with the Combo loss function, achieving a DSC score of 0.791 and an HD95 score of 21.66, outperforming other models. In addition, the density plot analysis demonstrated that with the proposed segmentation model, the distribution scores of all classes were greater than those with other models.

In the GA estimation task, different regression techniques (LR, RF, RT, and XGB) were assessed in combination with three feature extraction approaches (3D autoencoder, radiomics, and IRMMNET encoder). We found that when used with with RF, the radiomics features led to the best GA prediction results, with an RMSE score of 1.42 and a C-index score of 0.888. In addition, the analysis of the Kaplan–Meier plots proved the efficacy of the proposed GA prediction method based on the radiomics features and RF. Further, we studied the explainability and the importance of the radiomics features, and we found that the max 3D diameter and minor axis length were the most important features in GA prediction in comparison with the other features.

Finally, we studied the generalization capabilities of the proposed fetal brain segmentation and GA prediction methods for two tasks, namely, head and neck tumor segmentation and the prediction of patients' survival days. When applied to head and neck tumor segmentation, we found that the proposed segmentation model outperformed existing models that were specifically designed for this task. In addition, we found that the proposed features based on radiomics and deep features extracted from the IRMMNET encoder achieved higher C-index scores (0.821) than those of the methods that were specially designed for the prediction of patients' survival days.

The presented approach to segmenting the fetal brain and estimating gestational age and delivery dates could be utilized to accurately quantitatively analyze the human fetal brain and to determine the gestational age and expected delivery date, both of which are crucial for management decisions in any pregnancy. For the timing of appropriate obstetric care, scheduling, and the interpretation of several antepartum diagnostics, precise information on the gestational age is essential.

In future work, we will use different patch-based 3D segmentation models and 3D transformers for the task of fetal brain segmentation in order to further enhance the system's performance. In addition, we will validate the GA estimation method with data from larger populations and settings.

Author Contributions: Conceptualization, M.M., A.Q., D.P. and M.A.-N.; methodology, M.M., A.Q., D.P. and M.A.-N.; software, M.M.; validation, M.M.; formal analysis, M.M., A.Q., D.P. and M.A.-N.; investigation, M.M., A.Q., D.P. and M.A.-N.; resources, M.A.-N. and D.P.; data curation, M.M.; writing—original draft preparation, M.M., A.Q. and M.A.-N.; writing—review and editing, M.M., A.Q., D.P. and M.A.-N.; visualization, M.M., A.Q. and M.A.-N.; writing—review and editing, M.M., A.Q., D.P. and M.A.-N.; to supervision, D.P. and M.A.-N.; project administration, M.A.-N. and D.P.; funding acquisition, M.A.-N. and D.P. All authors have read and agreed to the published version of the manuscript.

Funding: The Spanish Government partly supported this research through Project PID2019-105789RB-I00.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** The datasets used in this study are publicly available at https://feta.grand-challenge.org/feta-2021/. The source codes of the proposed method will be available at https://github.com/Moona-Mazher/Fetal-Segmentation-Gestational-Age-Prediction-Deep-Learning on 20 November 2022.

Conflicts of Interest: The authors declare no conflict of interest.

# References

- Ebner, M.; Wang, G.; Li, W.; Aertsen, M.; Patel, P.A.; Aughwane, R.; Melbourne, A.; Doel, T.; Dymarkowski, S.; De Coppi, P.; et al. An automated framework for localization, segmentation and super-resolution reconstruction of fetal brain MRI. *NeuroImage* 2020, 206, 116324. [CrossRef] [PubMed]
- Egaña-Ugrinovic, G.; Sanz-Cortes, M.; Figueras, F.; Bargalló, N.; Gratacós, E. Differences in cortical development assessed by fetal MRI in late-onset intrauterine growth restriction. *Am. J. Obstet. Gynecol.* 2013, 209, 126.e1–126.e8. [CrossRef] [PubMed]
- 3. Jarvis, D.A.; Finney, C.R.; Griffiths, P.D. Normative volume measurements of the fetal intra-cranial compartments using 3D volume in utero MR imaging. *Eur. Radiol.* 2019, 29, 3488–3495. [CrossRef]
- Habas, P.A.; Kim, K.; Corbett-Detig, J.M.; Rousseau, F.; Glenn, O.A.; Barkovich, A.J.; Studholme, C. A spatiotemporal atlas of MR intensity, tissue probability and shape of the fetal brain with application to segmentation. *Neuroimage* 2010, 53, 460–470. [CrossRef]
- Falick Michaeli, T.; Spiro, A.; Sabag, O.; Karavani, G.; Yagel, S.; Eventov-Friedman, S.; Cedar, H.; Bergman, Y.; Gielchinsky, Y. Determining gestational age using genome methylation profile: A novel approach for fetal medicine. *Prenat. Diagn.* 2019, 39, 1005–1010. [CrossRef] [PubMed]
- Khalili, N.; Lessmann, N.; Turk, E.; Claessens, N.; de Heus, R.; Kolk, T.; Viergever, M.A.; Benders, M.J.; Išgum, I. Automatic brain tissue segmentation in fetal MRI using convolutional neural networks. *Magn. Reson. Imaging* 2019, 64, 77–89. [CrossRef]
- Faghihpirayesh, R.; Karimi, D.; Erdogmus, D.; Gholipour, A. Deep Learning Framework for Real-time Fetal Brain Segmentation in MRI. *arXiv* 2022, arXiv:2205.01675.

- Asis-Cruz, D.; Krishnamurthy, D.; Jose, C.; Cook, K.M.; Limperopoulos, C. FetalGAN: Automated Segmentation of Fetal Functional Brain MRI Using Deep Generative Adversarial Learning and Multi-Scale 3D U-Net. *Front. Neurosci.* 2022, 16, 887634. [CrossRef]
- 9. Zhao, L.; Asis-Cruz, J.; Feng, X.; Wu, Y.; Kapse, K.; Largent, A.; Quistorff, J.; Lopez, C.; Wu, D.; Qing, K.; et al. Automated 3D Fetal Brain Segmentation Using an Optimized Deep Learning Approach. *Am. J. Neuroradiol.* **2022**, *43*, 448–454. [CrossRef]
- Prasoon, A.; Petersen, K.; Igel, C.; Lauze, F.; Dam, E.; Nielsen, M. Deep feature learning for knee cartilage segmentation using a triplanar convolutional neural network. In Proceedings of the International Conference on Medical Image Computing and Computer-Assisted Intervention, Nagoya, Japan, 22–26 September 2013; pp. 246–253.
- 11. Chavez, M.R.; Ananth, C.V.; Kaminsky, L.M.; Smulian, J.C.; Yeo, L.; Vintzileos, A.M. Fetal transcerebellar diameter measurement for prediction of gestational age in twins. *Am. J. Obstet. Gynecol.* **2006**, *195*, 1596–1600. [CrossRef]
- 12. Wu, J.; Awate, S.; Licht, D.; Clouchoux, C.; Du Plessis, A.J.; Avants, B.; Vossough, A.; Gee, J.; Limperopoulos, C. Assessment of MRI-based automated fetal cerebral cortical folding measures in prediction of gestational age in the third trimester. *Am. J. Neuroradiol.* **2015**, *36*, 1369–1374. [CrossRef]
- Kyriakopoulou, V.; Vatansever, D.; Davidson, A.; Patkee, P.; Elkommos, S.; Chew, A.; Martinez-Biarge, M.; Hagberg, B.; Damodaram, M.; Allsop, J.; et al. Normative biometry of the fetal brain using magnetic resonance imaging. *Brain Struct. Funct.* 2017, 222, 2295–2307. [CrossRef] [PubMed]
- 14. Blondiaux, E.; Garel, C. Fetal cerebral imaging–ultrasound vs. MRI: An update. *Acta Radiol.* **2013**, *54*, 1046–1054. [CrossRef] [PubMed]
- 15. Namburete, A.I.; Stebbing, R.V.; Kemp, B.; Yaqub, M.; Papageorghiou, A.T.; Noble, J.A. Learning-based prediction of gestational age from ultrasound images of the fetal brain. *Med. Image Anal.* **2015**, *21*, 72–86. [CrossRef] [PubMed]
- Fung, R.; Villar, J.; Dashti, A.; Ismail, L.C.; Staines-Urias, E.; Ohuma, E.O.; Salomon, L.J.; Victora, C.G.; Barros, F.C.; Lambert, A.; et al. Achieving accurate estimates of fetal gestational age and personalised predictions of fetal growth based on data from an international prospective cohort study: A population-based machine learning study. *Lancet Digit. Health* 2020, 2, e368–e375. [CrossRef]
- 17. Kojita, Y.; Matsuo, H.; Kanda, T.; Nishio, M.; Sofue, K.; Nogami, M.; Kono, A.K.; Hori, M.; Murakami, T. Deep learning model for predicting gestational age after the first trimester using fetal MRI. *Eur. Radiol.* **2021**, *31*, 3775–3782. [CrossRef]
- Lu, Y.; Zhang, X.; Fu, X.; Chen, F.; Wong, K.K. Ensemble machine learning for estimating fetal weight at varying gestational age. In Proceedings of the AAAI Conference on Artificial Intelligence, Honolulu, HI, USA, 29–31 January 2019; Volume 33, pp. 9522–9527.
- 19. The Alliance for Maternal and Newborn Health Improvement (AMANHI) Gestational Age Study Group. Simplified models to assess newborn gestational age in low-middle income countries: Findings from a multicountry, prospective cohort study. *BMJ Glob. Health* **2021**, *6*, e005688. [CrossRef]
- Payette, K.; de Dumast, P.; Kebiri, H.; Ezhov, I.; Paetzold, J.C.; Shit, S.; Iqbal, A.; Khan, R.; Kottke, R.; Grehten, P.; et al. An automatic multi-tissue human fetal brain segmentation benchmark using the Fetal Tissue Annotation Dataset. *Sci. Data* 2021, *8*, 1–14. [CrossRef]
- 21. Shen, L.; Zheng, J.; Lee, E.H.; Shpanskaya, K.; McKenna, E.S.; Atluri, M.G.; Plasto, D.; Mitchell, C.; Lai, L.M.; Guimaraes, C.V.; et al. Attention-guided deep learning for gestational age prediction using fetal brain MRI. *Sci. Rep.* **2022**, *12*, 1408. [CrossRef]
- Szegedy, C.; Ioffe, S.; Vanhoucke, V.; Alemi, A.A. Inception-v4, inception-resnet and the impact of residual connections on learning. In Proceedings of the Thirty-First AAAI Conference on Artificial Intelligence, San Francisco, CA, USA, 4–9 February 2017.
- 23. Chen, L.C.; Papandreou, G.; Kokkinos, I.; Murphy, K.; Yuille, A.L. Deeplab: Semantic image segmentation with deep convolutional nets, atrous convolution, and fully connected crfs. *IEEE Trans. Pattern Anal. Mach. Intell.* **2017**, *40*, 834–848. [CrossRef]
- Woo, S.; Park, J.; Lee, J.Y.; Kweon, I.S. Cbam: Convolutional block attention module. In Proceedings of the European Conference on Computer Vision (ECCV), Munich, Germany, 8–14 September 2018; pp. 3–19.
- 25. Ramachandran, P.; Zoph, B.; Le, Q.V. Searching for activation functions. arXiv 2017, arXiv:1710.05941.
- 26. Taghanaki, S.A.; Zheng, Y.; Zhou, S.K.; Georgescu, B.; Sharma, P.; Xu, D.; Comaniciu, D.; Hamarneh, G. Combo loss: Handling input and output imbalance in multi-organ segmentation. *Comput. Med. Imaging Graph.* **2019**, *75*, 24–33. [CrossRef]
- Gopika, N.; Me, A.M.K. Correlation based feature selection algorithm for machine learning. In Proceedings of the 2018 3rd International Conference on Communication and Electronics Systems (ICCES), Coimbatore, India, 15–16 October 2018; pp. 692–695.
- Van Griethuysen, J.J.; Fedorov, A.; Parmar, C.; Hosny, A.; Aucoin, N.; Narayan, V.; Beets-Tan, R.G.; Fillion-Robin, J.C.; Pieper, S.; Aerts, H.J. Computational radiomics system to decode the radiographic phenotype. *Cancer Res.* 2017, 77, e104–e107. [CrossRef] [PubMed]
- Almalki, Y.E.; Qayyum, A.; Irfan, M.; Haider, N.; Glowacz, A.; Alshehri, F.M.; Alduraibi, S.K.; Alshamrani, K.; Alkhalik Basha, M.A.; Alduraibi, A.; et al. A novel method for COVID-19 diagnosis using artificial intelligence in chest X-ray images. *Healthcare* 2021, 9, 522. [CrossRef] [PubMed]
- Fernández-Delgado, M.; Sirsat, M.S.; Cernadas, E.; Alawadi, S.; Barro, S.; Febrero-Bande, M. An extensive experimental survey of regression methods. *Neural Netw.* 2019, 111, 11–34. [CrossRef] [PubMed]

- 31. Qayyum, A.; Lalande, A.; Meriaudeau, F. Automatic segmentation of tumors and affected organs in the abdomen using a 3D hybrid model for computed tomography imaging. *Comput. Biol. Med.* **2020**, *127*, 104097. [CrossRef]
- 32. Qayyum, A.; Lalande, A.; Decourselle, T.; Pommier, T.; Cochet, A.; Meriaudeau, F. Segmentation of the myocardium on late-gadolinium enhanced MRI based on 2.5 D residual squeeze and excitation deep learning model. *arXiv* **2020**, arXiv:2005.13643.
- Zabihollahy, F.; Rajchl, M.; White, J.A.; Ukwatta, E. Fully automated segmentation of left ventricular scar from 3D late gadolinium enhancement magnetic resonance imaging using a cascaded multi-planar U-Net (CMPU-Net). *Med Phys.* 2020, 47, 1645–1655. [CrossRef]
- de Dumast, P.; Kebiri, H.; Payette, K.; Jakab, A.; Lajous, H.; Cuadra, M.B. Synthetic magnetic resonance images for domain adaptation: Application to fetal brain tissue segmentation. In Proceedings of the 2022 IEEE 19th International Symposium on Biomedical Imaging (ISBI), Kolkata, India, 28–31 March 2022; pp. 1–5.
- Dumast, P.d.; Kebiri, H.; Atat, C.; Dunet, V.; Koob, M.; Cuadra, M.B. Segmentation of the cortical plate in fetal brain MRI with a topological loss. In Uncertainty for Safe Utilization of Machine Learning in Medical Imaging, and Perinatal Imaging, Placental and Preterm Image Analysis; Springer: Cham, Switzerland, 2021; pp. 200–209.
- 36. Antwarg, L.; Miller, R.M.; Shapira, B.; Rokach, L. Explaining anomalies detected by autoencoders using Shapley Additive Explanations. *Expert Syst. Appl.* **2021**, *186*, 115736. [CrossRef]
- 37. Andrearczyk, V.; Oreiller, V.; Boughdad, S.; Rest, C.C.L.; Elhalawani, H.; Jreige, M.; Prior, J.O.; Vallières, M.; Visvikis, D.; Hatt, M.; et al. Overview of the HECKTOR challenge at MICCAI 2021: Automatic head and neck tumor segmentation and outcome prediction in PET/CT images. In Proceedings of the 3D Head and Neck Tumor Segmentation in PET/CT Challenge, Strasbourg, France, 27 September 2021; pp. 1–37.
- 38. Bourigault, E.; McGowan, D.R.; Mehranian, A.; Papież, B.W. Multimodal PET/CT Tumour Segmentation and Prediction of Progression-Free Survival using a Full-Scale UNet with Attention. *arXiv* **2021**, arXiv:2111.03848.
- Starke, S.; Leger, S.; Zwanenburg, A.; Leger, K.; Lohaus, F.; Linge, A.; Schreiber, A.; Kalinauskaite, G.; Tinhofer, I.; Guberina, N.; et al. 2D and 3D convolutional neural networks for outcome modelling of locally advanced head and neck squamous cell carcinoma. *Sci. Rep.* 2020, *10*, 15625. [CrossRef] [PubMed]
- Aerts, H.J.; Velazquez, E.R.; Leijenaar, R.T.; Parmar, C.; Grossmann, P.; Carvalho, S.; Bussink, J.; Monshouwer, R.; Haibe-Kains, B.; Rietveld, D.; et al. Decoding tumour phenotype by noninvasive imaging using a quantitative radiomics approach. *Nat. Commun.* 2014, 5, 1–9. [CrossRef] [PubMed]
- Juanco-Müller, Á.V.; Mota, J.F.; Goatman, K.; Hoogendoorn, C. Deep supervoxel segmentation for survival analysis in head and neck cancer patients. In Proceedings of the 3D Head and Neck Tumor Segmentation in PET/CT Challenge, Strasbourg, France, 27 September 2021; pp. 257–265.